\documentclass[letterpaper]{article}
%\documentclass[twocolumn,letterpaper]{article}
\usepackage{vmargin,natbib}
%\usepackage{times,mathptmx,vmargin,apalike}
\usepackage{graphicx}
\usepackage{hyperref}
\hypersetup{colorlinks=true,filecolor=blue,urlcolor=blue,breaklinks=true}
%
\usepackage{epsfig}
\setpapersize{USletter}
\setmarginsrb{1in}{1in}{1in}{1in}{0pt}{0mm}{0pt}{6mm}
\setlength{\columnsep}{0.25in}
\setlength{\parskip}{\baselineskip}
\setlength{\parindent}{0pt}
%\renewcommand{\familydefault}{phv}
%%\DeclareSymbolFont{symbols}{OMS}{psyr}{m}{n}
%%\SetSymbolFont{symbols}{normal}{OMS}{psyr}{m}{n}
\begin{document}

\title{Forrest M. Hoffman}

\author{
Computational Earth Sciences Group \\
Oak Ridge National Laboratory (ORNL) \\
Oak Ridge, Tennessee 37831-6016 \\
865-576-7680 \\
\texttt{forrest@climatemodeling.org} \\
}

\date{}
\maketitle
\pagestyle{empty}

\section{Professional Preparation}

\begin{table}[h]
  \begin{tabular}{l l l}
University of Tennessee, Knoxville & Physics & B.S. 1991 \\
University of Tennessee, Knoxville & Physics & M.S. 2004 \\
  \end{tabular}
\end{table}

\section{Appointments}

\textbf{Oak Ridge National Laboratory (ORNL)/UT-Battelle, LLC}, MS 6016,
Oak Ridge, Tennessee 37831--6016: Develops and implements serial and
parallel scientific models used in environmental analyses and predictions.
Develops supporting software tools for visualization, analysis, and
publication of scientific data and information.  Performs systems analysis
and design for the Atmospheric Radiation Measurement (ARM) Archive.
Develops and impelements Internet technologies for the U.S. Global Change
Data and Information System (GCDIS), the U.S. National Assessment,
and Project NOPP Drifters.  Requires knowledge of Unix systems,
the X Window System, HTML, JavaScript, CGI, C, FORTRAN, computer and
network security, computer programming, and modeling and visualization
techniques.  Currently serving on the ORNL Scientific Computing User
Advisory Committee, the ORNL IBM SP2 Access Committee, the ORNL Internet
Technologies Working Group (ITWG), the ORNL Secure On-Line Facilities
Development Group, and the ESD Electronic Communications Committee
(11/1993--present).

\textbf{\textit{Linux Magazine}}: Contributing editor and columnist for
``Extreme Linux'' column (01/2002--10/2006).

\textbf{University of Tennessee, Department of Geological Sciences},
Knoxville, Tennessee 37996:  Performed scientific software development,
system administration, systems programming, and management of several
Unix-based workstations and minisupercomputers in the Earth Sciences
Modeling Center of Oak Ridge National Laboratory's Environmental Sciences
Division.  Developed and modified scientific models and tools used in
environmental analyses, including numerical models and visualization
systems used to model and analyze environmental data (7/1992--11/1993).

\textbf{Automated Sciences Group, Inc.}, 800 Oak Ridge Turnpike, Oak
Ridge, Tennessee 37830: Knowledge Engineer for Hydrogeochemical Expert
System Project at Oak Ridge National Laboratory.  Responsible for software
design and development for contaminant transport modeling and Unix system
administration and programming.  Proficiency in knowledge engineering,
knowledge representation, programming languages, and scientific modeling
techniques required (5/1989--6/1992).

\textbf{Oak Ridge Associated Universities (ORAU)}, Oak Ridge,
Tennessee 37830:  Science and Engineering Research and Professional
Intern for Hydrogeochemical Expert System Project at Oak Ridge National
Laboratory (9/1988--5/1989).

\textbf{National Center for Atmospheric Research}, Mauna Loa Solar
Observatory, Hilo, Hawai'i 96721--0425:  Operator of K-Coronameter,
Prominence Monitor, and Calcium Variability Monitor.  Performed scientific
data collection and management, operated multiple computer systems and
telescope control systems at high altitude observatory (1987--9/1988).


\nocite{
Kloster_Biogeosci_20100611,
Randerson_GCB_20091001,
Erickson_JGR_20080315,
Hoffman_JPConf_20071201,
Dickinson_JClim_20060601,
Hargrove_JGS_20060701,
Hoffman_IJHPCA_20050801,
Hoffman_EI_20050803,
Hargrove_LandscapeEcol_20050501,
White_GRL_20050218,
Saxon_EcolLett_20050101,
Hargrove_EnvironManage_20040401,
Hargrove_ConservEcol_20020213,
Hargrove_SciAm_20010801}

\bibliographystyle{unsrtnat}
\renewcommand\refname{Selected Publications}
\bibliography{abbrev,hoffman,linux-mag}

\end{document}
